

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers' comments concerning our manuscript entitled "**One-Stage Total Hip Arthroplasty for Advanced Hip Tuberculosis Combined with Developmental Dysplasia of the Hip: A Case Report**" (ID: 68347). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. The main corrections in the paper and the responds to the reviewer's comments are as following:

Responds to the reviewer's comments:

- (1) We have uploaded the Conflict-of-Interest Disclosure Form.
- (2) We have provided the PMID and DOI numbers are missing in the reference list.
- (3) All figures legends have been formatted in the uniform presentation.
- (4) The title of the manuscript has been shortened to 18 words.